A meta-analysis of factors influencing the strength of mate choice copying in animals

Alice D. Davies¹, Zenobia Lewis² & Liam R. Dougherty^{1*}

¹Institute of Integrative Biology, University of Liverpool, Liverpool, L69 7ZB, UK ²School of Life Sciences, University of Liverpool, Liverpool, L69 7ZB, UK

*Corresponding author email: liam.dougherty@liverpool.ac.uk

Funding

This work was supported by a Leverhulme Trust Early-Career Fellowship to LRD (ECF-2018-427).

Acknowledgements

We would like to thank Rachel Moran, Sabine Nöbel and Klaudia Witte for sending us their data, and two anonymous reviewers for comments that greatly improved the manuscript.

Lay summary

Meta-analysis shows that both male and female animals are more likely to copy, rather than avoid, the mate choices of their rivals. We analysed the results of 58 studies testing how animal mate choice is influenced by seeing the choice of a rival. We found that animals do copy the choices of rivals, though copying was strongest in mammals and weakest in invertebrates. Both males and females copied to a similar degree.

Abstract

Mate-choice copying is a form of social learning in which an individual's choice of mate is influenced by the apparent choices of other individuals of the same sex, and has been observed in more than 20 species across a broad taxonomic range. Though fitness benefits of copying have proven difficult to measure, theory suggests that copying should not be beneficial for all species or contexts. However, the factors influencing the evolution and expression of copying have proven difficult to resolve. We systematically searched the literature for studies of mate-choice copying in non-human animals, and then performed a phylogenetically-controlled meta-analysis to explore which factors influence the expression of copying across species. Across 58 published studies in 23 species, we find strong evidence that animals copy the mate choice of others. The strength of copying was significantly influenced by taxonomic group, however sample size limitations mean it is difficult to draw firm conclusions regarding copying in mammals and arthropods. The strength of copying was also influenced by experimental design: copying was stronger when choosers were tested before and after witnessing a conspecific's mate choice, compared to when choosers with social information were compared to choosers without. Importantly, we did not detect any difference in the strength of copying between males and females, or in relation to the rate of multiple mating. Our search also highlights that more empirical work is needed to investigate copying in a broader range of species, especially those with differing mating systems and levels of reproductive investment.

Keywords

Mate choice, mate choice copying, non-independent mate choice, learning, meta-analysis

Introduction

In the natural environment, individuals vary in their quality as a mate, in terms of the resources they can provide to mates, and the parental care or genes they can provide to offspring (Kokko et al., 2003). Because of this, there are often large fitness benefits to choosing the highest-quality mates available via the process of mate choice (Rosenthal, 2017). In social species, interactions between males and females may often be visible to other members of the population. This means that mating, or any other affiliative behaviour, has the potential to create public information which others can use. This information is transmitted inadvertently, making it typically an honest and reliable signal of mate quality (Vakirtzis, 2011). In species with the cognitive ability to recognise and remember such information, there is the potential for observers to use the choices of other individuals of the same sex (referred to as 'demonstrators' or 'models') to inform their own mating decisions, a process known as mate-choice copying (Pruett-Jones, 1992). Mate-choice copying is a form of social learning in which an individual's choice of mate is influenced by the apparent choices of other conspecifics of the same sex (Wade & Pruett-Jones, 1990; Dugatkin, 1992). Typically, the attractiveness of a potential partner (which we call the 'target' individual) is increased after observing a rival demonstrator interacting positively with the partner. Copying is predicted to reduce the cost associated with sampling mates (Wade & Pruett-Jones, 1990; Vakirtzis, 2011; Dubois et al., 2011), and may be especially advantageous when quality is difficult to assess (Mery et al., 2009; Vakirtzis, 2011) or the copier is inexperienced (Dugatkin and Godin, 1993). Conversely, in some cases copying could lead to poorer choices, if copying causes observers to ignore signals of low quality (Dubois et al., 2011). For example, male guppies (*Poecilia reticulata*) usually prefer larger, more fecund females,

but in the presence of social information can be led to prefer the smaller of two females (Auld and Godin, 2015).

Theoretically, any species which exhibits mate choice and possesses the cognitive capacity for social learning (Shettleworth, 2010) could exhibit mate-choice copying. To date, matechoice copying has been recorded in over 20 non-human species, including mammals, birds, fishes, spiders, and insects (Vakirtzis, 2011; Jones & DuVal, 2019). Even species commonly believed to have minimal cognitive abilities, such as the fruit fly Drosophila melanogaster, have repeatedly been shown to exhibit mate-choice copying (Mery et al., 2009; Danchin et al., 2018). A subset of studies also find evidence for 'generalised' copying, whereby observers apply a newly-observed preference to all mates with the same phenotype as the chosen target individual (Vakirtzis, 2011). This has been seen in fish (Godin et al., 2005; Witte & Noltemeier, 2002), birds (White & Galef, 2000a; Swaddle et al., 2005), and arthropods (Mery et al., 2009; Fowler-Finn et al., 2015). However, a growing number of studies have tested for mate-choice copying and found a non-significant effect (e.g. Lafleur et al., 1997; Auld et al., 2009; Milner et al., 2011). Further, in some species copying has been found in some tests but not others (e.g. guppies P. reticulata: Dugatkin, 1992, Lafleur et al., 1997; zebra finches Taeniopygia guttata: Doucet et al., 2004, Swaddle et al., 2005). Indeed, a recent meta-analysis has shown quantitatively that the prevalence and strength of copying varies greatly both across and within species (Jones & DuVal, 2019).

This raises the question: what factors influence the evolution or expression of copying? The fitness benefits of copying have proven difficult to measure (Witte et al., 2015), but theory nevertheless predicts that copying will not be advantageous in all contexts and for all species. For example, copying is predicted to increase copier fitness when mate sampling is costly,

quality is difficult to assess, and preferences take time to learn, and should be more likely in species with high sociality and overlapping generations (Varela et al., 2018). Additionally, species which are strictly monogamous, or require biparental care for the successful rearing of offspring, should be less likely to exhibit individual mate-choice copying, because of the high costs associated with sharing mates (Vakirtzis and Roberts, 2009; though note that this cost does not apply to generalised copying). In many cases monogamous species do not copy, as predicted (e.g. Slagsvold and Viljugrein, 1999). However, in other cases, mate-choice copying has been experimentally observed in a number of socially or exclusively monogamous species. These include the monandrous wolf spiders of the genus Schizocosa (Norton and Uetz, 2005, Fowler-Finn et al., 2015) and the socially monogamous Japanese quail Coturnix japonica (Nichols, 1991). The occurrence of mate-choice copying could also be affected by factors relating to experimental design, either because this influences our ability to measure copying, or because the behaviour of copiers differs between experiments (Dougherty, 2020). For example, studies may also differ in the extent to which the demonstrator is able to interact with the target. Mate-choice copying may occur in response to the observation of mating (e.g. Fowler-Finn et al. 2015, Nöbel et al., 2018a) or close affiliation (e.g. Dugatkin and Godin, 1992, Gierzewski et al., 2018) between demonstrators and targets. Given that mating is likely a stronger signal of mate quality than sexual interest or affiliation, observers may be more likely to copy after witnessing an actual mating event compared to the latter behaviours.

The sex of the observer is another factor which is often suggested to influence copying (Witte et al., 2015). Females are assumed to be more likely to copy than males, because the loss of fitness incurred by making an incorrect assessment of mate quality is greater for females, who typically invest more into reproduction than males (Trivers, 1972). Sharing a mate may

also be more costly for males, because sperm competition reduces male fitness when mating with non-virgin females (Simmons, 2001; Vakirtzis and Roberts, 2009). Nevertheless, evidence for male mate-choice copying has been found in several species (e.g. Schlupp and Ryan, 1997; Wong and McCarthy, 2009; Auld and Godin, 2015; Nöbel et al., 2018a; but see White and Galef, 2000b, Widemo, 2006), though it is unclear whether it is generally weaker than female copying. Notably, male copying in several species may be explained by the fact that males may invest in reproduction as much as, or even more than, females (Widemo, 2006; Moran et al., 2013; Witte et al., 2015; Scauzillo & Ferkin, 2019).

Overall then, there are now a large number of studies that have examined mate-choice copying in animals, and which have shown that the prevalence and strength of copying varies greatly across and even within species. However, the factors driving this variation remain largely unclear. Though traditional reviews can highlight promising areas for study (e.g. Vakirtzis, 2011; Witte et al., 2015; Scauzillo & Ferkin, 2019), a focus on significant or non-significant P values can obscure more subtle patterns (Koricheva et al., 2013). By converting the results of these studies into a standardised effect size (rather than *P* values), we can use meta-analysis to test how a range of biological and methodological factors influence the direction (i.e. whether observers copy or avoid the choices of demonstrators) and strength of mate-choice copying. This method was recently used to show that female mate-choice copying is tested in free-living compared to captive subjects, and when demonstrator choice is counter to the observer's preference (Jones & DuVal, 2019).

Here we present the results of a phylogenetically-controlled meta-analysis examining factors influencing the expression of mate-choice copying across animals. We systematically

surveyed the literature to obtain as many studies as possible examining mate-choice copying in both males and females, across all species excluding humans. This resulted in 158 effect sizes, from 58 studies and 23 species. We then formally test whether a range of biological and methodological factors influence the strength of copying across species. Though our searches resulted in a similar dataset to that used in the recent meta-analysis by Jones & DuVal (2019; and our searches were performed within a month of each other), our study differs in two important ways. First, we explicitly include studies examining mate-choice copying in males, in order to compare the degree of copying between males and females across species. Second, we test the effects of several biological and methodological factors not considered by Jones & DuVal (2019). Taken together, both studies provide a quantitative picture of the current evidence for mate-choice copying across animals, and begin to identify factors which influence the strength of copying.

Methods

Literature search and inclusion criteria

We searched for relevant studies using two main approaches. First, we collected all empirical examples cited in the review of mate-choice copying by Vakirtzis (2011), in addition to all papers citing this review up to the 4th April 2019. Second, we searched the online databases Web of Science and Scopus (on the 4th April 2019) for studies containing the following search terms: "mate choice" AND (copying OR non-independent OR nonindependent OR learning), across all years. We used the 'Topic' search field in Web of Science and the 'Article Title, Abstract, Keywords' search field in Scopus. This returned 642 results from Web of Science and 552 from Scopus, yielding a total of 1230 papers, reduced to 844 after the removal of duplicates (Figure 1). The titles and abstracts of these papers were then

examined to determine whether they met our inclusion criteria (see below). The numbers of papers rejected, and reasons why, are given in Figure 1. This sorting left 89 papers which were read in their entirety.

A study had to meet several criteria for it to be included in our analysis. First, it must test for mate-choice copying in either females or males. We considered an experiment to test for mate-choice copying if the sexual preference of a focal individual was compared in the presence and absence of social information regarding the preference of a 'demonstrator' individual of the same sex. This definition includes studies that do not directly test sexual preferences in the absence of social information, but instead compare post-demonstration preferences to hypothetical values (see below). We excluded studies that were purely observational, did not provide the observer with direct social information (e.g. demonstrator choices inferred from olfactory cues in Kavaliers et al., 2006), or did not compare behaviour to approportiate control data or an appropriate null hypothesis. Strictly-speaking, copying a demonstrator preference could increase or decrease the attractiveness of the target individual (or phenotype), depending on whether the observer witnessed acceptance or rejection by the demonstrator (Pruett-Jones, 1992; Vakirtzis, 2011; Scauzillo & Ferkin, 2019). However, in practice copying of rejection is rarely tested (Scauzillo & Ferkin, 2019), and only one study in our sample tested for an effect of mate rejection on observer behaviour in isolation (Witte & Ueding, 2003). We excluded studies examining forms of learning other than copying, such as social imprinting of sexual preferences (e.g. Bailey & Zuk, 2009). Second, studies needed to provide sufficient data or statistical information to allow us to calculate a standardised effect size. When such information was missing (e.g. missing sample sizes or test statistics), we attempted to contact the authors, requesting either the missing information or the raw data. We obtained the raw data for five studies in this way. Within these limitations, we

considered studies using a variety of experimental designs (see below). We considered studies in any animal species, excluding humans (for a recent review and meta-analysis of mate-choice copying in humans, see Gouda-Vossos et al., 2018). We considered copying by both males and females. A list of excluded papers, and their reasons for exclusion, is presented in Table S1.

[Fig 1 here]

Effect size calculations

We quantified mate-choice copying as the difference in mating preference in the presence versus absence of social information (social versus control treatment). The greater the difference in preference between treatments, the greater the influence of social information on mate choice. We used the standardised mean difference, known as Hedges' *d*, as our measure of effect size (Hedges and Olkin, 1985). This effect size is useful because it is easily obtained using a range of different data types (Koricheva et al., 2013). In studies where the focal individual behaved in the same way as the demonstrator, known as 'positive copying', the effect size was given a positive sign. In studies where the focal individual behaved differently to the demonstrator (e.g. when an observer avoids the mate chosen by the demonstrator), known as 'negative copying', the effect size was given a negative sign. All data were extracted by one person (ADD). We extracted all relevant effect sizes from each study. In some cases this resulted in multiple effect sizes for a given study, typically because studies contain multiple experiments or report several behaviours. This potential source of non-independence was controlled for in the analysis (see below).

Effect sizes were calculated from two types of data. First, we used counts of the number of choices made towards each option (2 x 2 contingency table). In these cases we either used chi-squared statistics presented in the text, or performed our own chi square goodness-of-fit tests. The results of chi-squared tests were then converted into d using the equation in Koricheva et al. (2013). Second, we used pairs of means representing the average preference in social and control treatments, along with some measure of variance. In this case d can be calculated directly using means, standard deviations, and sample sizes for each treatment group. Means and standard deviations were taken directly from the text, or extracted from figures using Web Plot Digitizer v 4.2 (Rohatgi, 2019). Alternatively, the test statistics from t-tests or paired t-tests were converted into d using the equations in Koricheva et al. (2013). Preferences were typically tested using a dichotomous-choice experimental design (Dougherty & Shuker, 2015), resulting in two preference values per test. In these cases we compared the average preference for the option chosen by the demonstrator between the two treatments, or used a difference score if provided by the authors (the difference in association time between the most-preferred and least-preferred options). For preference scores originating from repeated-measures designs (27 effect sizes), the correlation between pre and post-test preferences were generally not reported, and so we assumed a correlation of 0.5 in order to convert paired-t statistics into d. This will underestimate the true effect size slightly if correlations between preferences are typically larger than this (though across species the repeatability of mating preferences is generally low: Bell et al., 2009). For non-parametric data, we converted the z score from a Wilcoxon test (either paired or non-paired) to d using the equation in Koricheva et al. (2013). For both types of data, social treatment groups were compared either to control groups, or with a null prediction of random mating (an equal frequency of mating with each choice option, or a difference score of zero) unless other theoretical 'expected' values were specified.

We set the direction of the effect size (positive or negative) following the rules stated above. We also obtained 13 effect sizes with no direction, for two reasons. First, count data can result in a 'true' effect size of zero if the frequency of each option does not differ between tests. Second, some studies reported the result of a non-significant statistical test, but provided insufficient information for us to estimate the direction of the effect (e.g. no means provided). These results are typically excluded from meta-analyses (Harts et al., 2016), however this biases the data against non-significant results, which are equally as informative as significant ones. We therefore assigned these effect sizes a value of zero and included them in all analyses, assuming that there is an approximately equal chance that any given result is positive or negative. This is likely conservative, given the fact that positive effect sizes are more common in this dataset (see below). We did this for seven cases.

After the completion of our data collection and statistical analysis the meta-analysis by Jones & DuVal (2019) was published. Coincidentally, the literature searches for the two studies were performed at almost the same time (April and May 2019). This provides the opportunity to assess how, given the same pool of available studies, different research teams may pursue different inclusion criteria and analytical approaches. This is important because meta-analyses may present researchers with an especially large number of subjective choices ("researcher degrees of freedom": Wicherts et al, 2016; Lakens et al., 2016), which may contribute to cases where two groups have analysed the same literature and got different or conflicting results (e.g. Gildersleeve et al, 2014; Wood et al, 2014; Silberzahn et al, 2015). For illustration, we therefore compare our methodology to that of by Jones & DuVal (2019) in the supplementary material (Table S2).

Moderators and accompanying hypotheses

One aim of this study was to determine which biological and methodological factors influence the direction or strength of copying. For each effect size, we obtained information on the following nine moderator variables:

Biological moderators:

Taxonomic group. The degree of mate-choice copying may differ across taxonomic groups, due to assumed differences in cognitive capacity or sociality. Mate-choice copying has been most commonly investigated in fish and birds, with some tests in mammals and arthropods (including crustaceans, arachnids, and insects). If cognitive capacity is important, we predict that mammals and birds will show the strongest degree of copying, followed by fishes and then arthropods.

Sex. Though copying has been recorded for both males and females, across species we expect copying to be stronger in females, for two reasons. First, females typically invest more in reproduction, and have lower mating rates than males, and so they are under greater selective pressure to make accurate judgements of quality (Mery et al., 2009, Vakirtzis, 2011). Second, males may suffer reduced fertilisation success when mating with recently-mated females due to sperm competition (Simmons, 2001), so that in many species males should actively avoid mating with such females.

Rate of multiple mating. The rate of multiple mating has the potential to influence the costs and benefits of copying, due to effects on the number of future mating opportunities and the degree of competition between rivals. If mating rates are high, copying should be lower because each mating event is less important, and so individuals can be less choosy overall.

Further, both males and females may suffer costs when mating with already-mated members of the opposite sex: from a female perspective, multiply-mated males may be sperm limited (Wedell et al., 2002) or provide less parental resources if they have another brood to care for (Vakirtzis and Roberts, 2009); from a male perspective, multiply-mated females are a high sperm competition risk (Simmons, 2001). We searched the literature for relevant data in order to categorise each species as having either a high or low rate of multiple mating (Table S3). In the first instance we used observations of the number of matings per individual or genetic paternity data to classify species. We classed a species as having a high rate of multiple mating if the majority of broods (>50%) were of mixed paternity, or individuals were observed to mate with an average of two or more partners during a breeding season. For three species we were unable to obtain quantitative estimates of mating rate, but we did find descriptions of the mating system which allowed us to assign species to one of the categories (Table S3). Two species could not be categorised in this way due to a lack of information (the humpbacked limia Limia nigrofasciata and the Atlantic molly Poecilia mexicana), and mating rate is not relevant to the parthenogenetic Amazon molly *Poecilia formosa* (Schartl et al., 1995), so all three were excluded from the analysis when testing for the effect of this moderator (this applies to nine effect sizes in total).

Animal birth location. The recent meta-analysis by Jones & DuVal (2019) found that copying was significantly stronger in field studies compared to those performed on captive individuals in the lab. This could be because mate sampling is more difficult in the wild, for example because there are more options to choose from, or because the risk of mortality is higher (Jones & DuVal, 2019). However, field studies also have less control over the experience of observers before testing, which could influence both overall mating preferences and the motivation to copy. In order to try to disentangle the effects of previous versus current testing

conditions on copying, we recorded whether individuals used during experiments were born in captivity or in the wild. The birth location has the potential to influence several factors that could influence the strength of copying, such as observer state or social experience (Cotton et al., 2006), and the costs of being choosy (Jennions & Petrie, 1997). We predict that copying will be weaker in wild-caught compared to lab-reared observers, as in the latter category more of these potentially-confounding environmental factors are controlled.

Experimental design moderators:

Individual vs generalised copying. Studies of mate-choice copying typically test how the attractiveness of a specific target individual changes following an observed sexual interaction with a demonstrator (individual copying). Alternatively, copying can lead to the formation of a generalised preference for a specific trait (generalised copying); this is confirmed by testing observer preferences for a different individual, with the same phenotype as the target. Distinguishing between these two types of copying is important, because they may lead to different ecological or evolutionary outcomes. For example, because a generalised rule can be applied to multiple individuals in a population, generalised copying has a greater potential to lead to widespread cultural transmission of preferences (Vakirtzis, 2011; Danchin et al., 2018). Generalised copying could also potentially aid the spread of novel ornaments in a population (e.g. Mery et al., 2009; Drullion and Dubois, 2008). However, it is not clear whether one type of copying is likely to be stronger than the other. Nevertheless, the two types of copying may require different learning abilities: individual copying requires the ability to recognise and remember specific individuals, whereas generalised copying requires the ability to apply a generalised rule to future mate encounters. It is also noteworthy that few studies have tested for generalised copying in males (Witte et al., 2015; Kniel et al., 2015), and only one of these studies found a significant result (Nöbel et al., 2018a).

Design type. We classified designs as using either a 'before-and-after' or a 'no pre-test' design. Before-and-after designs test observer preferences twice: once before demonstration and once after demonstration. No pre-test designs test for observer preferences once, following demonstration, and then compare this either to predicted values (e.g. random choice of 50%) or to individuals from a control group that did not witness a demonstration. A difference in copying between these two designs may arise due to the different assumptions of the two test designs, rather than a difference in observer behaviour. We predict that copying will be stronger using 'no pre-test' designs when studies compare observed preferences to predicted values. This is because the common default assumption of random choice will be incorrect if the two mate options differ even slightly in attractiveness. This would underestimate the true preference for the most attractive of the pair in the absence of social information, thus overestimating the difference in preferences between control and social treatments.

Model pairing. This category distinguishes between studies in which demonstrators are placed with an individual by the experimenters ('forced choice'), compared to studies in which demonstrators were genuinely able to choose between potential mates prior to the demonstration period ('actual choice'). We predicted that copying should be strongest in the actual choice category, because demonstrators here may exhibit subtle, additional behavioural signals of preference which are absent in the forced choice category.

Demonstrator behaviour. This category compared how the demonstrator interacts with the stimulus mate: is actual mating allowed to take place, or do individuals just affiliate with each other? Intuitively, observing mating rather than affiliation should provide a stronger social

signal of the demonstrator's interest in mating, and so we predicted this should lead to stronger copying.

Preference measure. How is the observer preference measured? The type of measurement has the potential to influence the strength of mating preferences for two reasons (Rosenthal, 2017; Dougherty, 2020). First, the measurement of choice may influence how observers behave during tests. For example, recording mating outcomes could result in weaker preferences compared to a proxy behavioural measure such as affiliation, because mating requires an interaction between the sexes, or because affiliation can be confounded by the time spent assessing partners (Rosenthal, 2017). Second, preference measures differ statistically in ways that could influence the observed effect size. For example, studies differ in whether preference is determined using some continuous measure such as association time, or a dichotomous measure of choice. If binary choice measures are derived from some continuous measure, then this conversion could potentially mask variation, or over-estimate preference differences because even small differences in preference are converted into an allor-nothing response (Dougherty, 2020). This could be especially problematic when choice is based on a simple majority (over 50%) criteria, as this may often be reached by chance. Conversely, studies that record the frequency of choice reversals may underestimate preference differences, because choice reversal is such a strict criteria: the observer needs to have an initial preference for one target which is then reversed following observation.

Phylogeny

Our sample included species distributed across several taxonomic groups, as well as some genera with multiple species represented (e.g. Poeciliidae). Therefore, we needed to control for non-independence associated with shared evolutionary history by obtaining a phylogeny

for the species sampled (Koricheva et al., 2013). As no single phylogeny was available which contained all of the species in our sample, we created a supertree combining phylogenetic and taxonomic information (Nakagawa & Santos, 2012). We created the tree using the R Open Tree of Life package v3.0.9 (Michonneau et al., 2016), with the positions of the genera *Poecilia, Gambusia* and *Limia* additionally determined using published phylogenies by Schartl et al. (1995), Ptacek and Breden (1998) and Reznick et al. (2017), respectively. As accurate branch lengths could not be obtained for the tree, branch lengths were initially set at 1. The tree was then made ultrametric, and branch lengths estimated, using Grafen's method (Grafen, 1989), and the R package Analysis of Phylogenetics and Evolution v 5.1 (Paradis et al., 2004). The final phylogenetic tree is shown in figure 2.

[Fig 2 here]

Statistical analysis

All analyses were performed in R version 3.5 (R Development Core Team, 2018). For metaanalysis we used the package Metafor v2.0 (Viechtbauer, 2010). We ran multilevel metaanalysis models using the rma.mv function, incorporating study identity, species, phylogeny (using the tree above), and observation ID as random factors. Study identity and species were included as we extracted multiple effect sizes per study, and collected data from more than one study for 8 species. Phylogeny was incorporated into all models using a variancecovariance matrix. Meta-analysis models require that observation ID be added as a random effect in order to estimate residual error.

We first calculated the overall mean effect size by running a model with just the four random factors listing above. We considered an effect size to significantly differ from zero when the

95% confidence intervals do not overlap zero. In order to compare our results to those of Jones & DuVal (2019), we converted our effect size into the odds ratio following Borenstein et al. (2009). We calculated the level of heterogeneity across all effect sizes using the I^2 statistic (Higgins et al., 2003). We also partitioned heterogeneity with respect to each of the four random factors, using the method of Nakagawa & Santos (2012). I^2 values of 25, 50 and 75% are considered low, medium and high respectively (Higgins et al., 2003).

To examine the effect of our moderator variables on the mean effect size, we ran mixedeffects multilevel meta-analytic models (also referred to as meta-regressions; Nakagawa & Santos, 2012). Each model included study identity, species, phylogeny and observation ID as random factors as before, but now also included one of the nine categorical moderator variables as a fixed effect. We note that this approach could potentially increase the chances of obtaining false positive results due to multiple testing. However, combining all of these fixed factors into a single model could potentially obscure significant effects, due to the modest sample size and potential collinearity among factors. We therefore take the former approach, and emphasise that statistically-significant effects should be interpreted with caution. We used the Q_M statistic to determine whether the mean effect size differed significantly across the different moderator categories (Koricheva et al., 2013). We also ran intercept-only mixed-effects models for each moderator variable separately, in order to obtain estimates of the mean effect size for each categorical factor level. For one of the categorical moderators, multiple mating rate, we obtained data for 20 out of 23 species, and so tested this fixed effect using a slightly reduced dataset (149 effect sizes). This also required the creation of a pruned tree with the three unclassified species removed.

We searched for signs of publication bias in our dataset in several ways. First, we searched

for changes in the average effect size over time. Effect sizes in ecology and evolution often decrease over time, an effect which is likely driven by publication practices: the earliest studies on a given topic tend to have large effects, and weak or non-significant results may often take longer to be published (Jennions & Møller, 2002). To examine this, we ran a metaregression with study year as a fixed effect, again with study identity, species and phylogeny as random effects. We examined the effect of study year on effect size using the Q_M statistic as described above. Second, we searched for signs of funnel plot asymmetry using two methods: Egger's test (Egger et al., 1997) and trim-and-fill (Duval & Tweedie, 2000). Asymmetry tests assume that effect sizes should be distributed randomly around the mean effect size; therefore, any deviation from symmetry could suggest some effect sizes are less likely to be published than others, especially when potential 'missing' effect sizes are of low precision and magnitude (a 'small study effect': Koricheva et al,. 2013). Egger's test is a regression of effect size against study precision, with asymmetry indicated by a slope that differs from zero. The trim-and-fill method imputes 'missing' effect sizes in order to make the funnel plot asymmetric. However, asymmetry tests are not appropriate when effect sizes are not independent, and it has been suggested that meta-analytic residuals are more appropriate (Nakagawa & Santos, 2012). We therefore obtained meta-analytic residuals using the MCMCglmm v2.25 package (Hadfield, 2010). MCMCglmm models used the same model structure as above (study ID, species, phylogeny and observation ID as random factors), and were run using an inverse-Wishart prior for all random effects, with models run for 300,000 iterations, a thinning interval of 50 and a burn-in period of 200,000 iterations. We performed both asymmetry tests using the residuals. However, we note here that asymmetry tests are expected to be less reliable when there are potential systematic causes of heterogeneity in the dataset other than publication bias, as the assumption of a single underlying distribution of

effect sizes is clearly false (Koricheva et al., 2013). As such, results of these tests should be interpreted with caution.

Results

The final dataset included 158 effect sizes from 58 studies and 23 species, across four taxonomic groups (Figure 2). The number of effect sizes per study ranged from one to 13, with an average of 2.8. Some species were very commonly studied; 61% of effect sizes came from the three most commonly studies species: the guppy *P. reticulata*, the sailfin molly *P. latipinna*, and the fruit fly *D. melanogaster* (Figure 3). We obtained 149 effect sizes (from 22 species) testing female copying, and 14 effect sizes (from 9 species) testing male copying (Figure 3). In almost all cases (eight out of nine species), our estimates of copying in males came from species for which we also had an estimate of copying for females.

[Fig 3 here]

Across all species, observation of a positive sexual interaction resulted in a significantly increased mating preference for the target individual or phenotype, thus providing strong evidence for positive mate-choice copying (Mean= 0.58, 95% confidence interval= 0.34 - 0.83; Figure 4). This difference corresponds to an odd ratio of 2.83 (95% CI= 1.85- 4.35). Few studies found evidence for avoidance of potential mates after observation of a positive sexual interaction between a target and a rival (only 24 negative effect sizes). The total amount of heterogeneity across effect sizes (I²) was high, at 84.82% (Figure 4). Partitioning heterogeneity among the three random factors indicated that most of the variance was at the observation level: 16.7% of heterogeneity was attributable to between-study differences, 0%

to between-species differences, 5.6% to phylogenetic history, and 62.47% to observationlevel differences.

[Fig 4 here]

The degree of mate-choice copying was significantly influenced by the taxonomic group of the study species and the type of experimental design (Table 1). In terms of taxonomic group, mammals appear to copy to the greatest extent, followed by fishes and birds, and then arthropods (Table S4; Figure 5). In terms of design type, studies using a before-and-after design detected more copying than studies with no pre-test (Table S4; Figure 5). The remaining seven categorical moderators did not significantly influence the degree of copying (Table 1; Table S4).

[Table 1 here]

[Fig 5 here]

There was no significant relationship between study year and effect size (Figure S1). Therefore there is no evidence for significant publication bias against low-powered, nonsignificant results when the research field was young. There were some signs of funnel plot asymmetry: trim-and-fill analysis (using residual effect sizes) detected 28 'missing' effect sizes, however all are positive and greater than the mean (Figure 6). This is likely driven by the large number of effect sizes close to zero, and to the left of the overall mean estimate (Figure 4). Therefore inclusion of these effect sizes increases the average effect size. The relationship between residual effect size and study precision was significantly negative (Egger's regression: F $_{1, 156}$ = 7.66, P= 0.006, β = -0.13; Figure S2), though it is unclear if this is driven by missing effect sizes of small effect and low power (which would be suggestive of publication bias), or of large effect and high power (as found by the trim-and-fill method).

[Fig 6 here]

Discussion

We performed a meta-analysis of 158 effect sizes from 58 published studies of mate-choice copying in non-human animals. Overall, we find strong evidence that animals copy the mating preferences of others. The overall standardised mean difference was 0.57 (95% CI= 0.34-0.81), which is considered a medium effect (Cohen, 1992). This converts to a mean odds ratio of 2.83, so that the observers in this sample are almost 3 times as likely to choose a partner or phenotype that they previously observed in a positive sexual interaction with a rival, compared to individuals with no social information. This result suggests that social information can have a significant influence on animal mating preferences across a wide range of species. Further, the strength of copying observed is similar to that reported in the recent meta-analysis of female copying by Jones & DuVal (2019: a mean odds ratio of 2.71), of which our dataset overlaps. We also detected significant variation in the degree of copying across species, which was partly explained by taxonomic group: copying was strongest for mammals, and weakest for arthropods. The strength of copying was also influenced by the type of experiment used to test for copying: copying was stronger when studies compared mating behaviour before and after the presentation of social information, in contrast to studies in which separate control and social treatment groups were compared. Interestingly, we did not detect any difference in copying between males and females, or in relation to the rate of multiple mating. We also detected little evidence for publication bias against non-significant

results.

While evidence suggests that mate-choice copying is taxonomically widespread, a small number of model species are over-represented in our dataset, and in the literature more generally: of the 89 studies identified in our search, 20 focus on guppies, nine on Japanese quail and nine on *D. melanogaster*. Despite this, species identity explained a small proportion of the variation in effect size in our analysis. This confirms that copying is not driven by a few influential species, and that within-species variability in effect size is high. Phylogenetic history explained only a small amount of variation in effect sizes, indicating that any similarities between species are not due to them being closely related evolutionarily. This is unsurprising, given the large evolutionary distances between the species in the dataset (with the exception of species in the Poeciliidae), and the evolutionary lability of behavioural traits (Blomberg et al., 2003). Nevertheless, we did detect differences in the degree of copying when we sorted the species in our dataset into four broad taxonomic groups: mammals, birds, fishes, and arthropods (incorporating insects, arachnids and crustaceans). The degree of copying was highest for mammals, followed by birds and fishes, and then lowest for arthropods. This finding could be explained by broad-scale differences in cognitive ability, ecological conditions or social group size. However, these results should be interpreted with caution: firstly because the number of effect sizes for mammals and arthropods was small, and secondly because the small number of species tested across the entire dataset means it is unclear how generalizable these results are. Importantly, we need more tests of copying in species groups other than those highlighted here. For example, to our knowledge there have been no experimental tests of copying in amphibians, reptiles, or non-human primates.

We found that the type of experimental design significantly influenced the degree of copying.

Copying was stronger for studies with a 'before-and-after' design, compared to studies with separate control and social treatment groups (no pre-test). This is the opposite of what we predicted, based on our hypothesis that no pre-test studies may often underestimate preferences in the absence of social information. We note however that our categories of 'before-and-after' and 'no pre-test' designs mostly align with the 'random' and 'unattractive' categories used by Jones and Duval (2019), which relate to whether the study used a reversal paradigm to test for copying (the unattractive category). The alignment comes from the fact that the reversal paradigm is a subset of the 'before-and-after' test. Accordingly, Jones and DuVal (2019) found that copying was stronger when using a reversal paradigm, which is consistent with our result shown here. One potential explanation for the observed effect could be related to the fact that before-and-after designs test each individual twice, potentially in quick succession. This provides the opportunity for observers to gain experience that could influence their choice in a non-random way. For example, individuals may become choosier with successive encounters because they have more information on the quality of mates in the population, they perceive mate density to be higher, or because they are attempting to 'tradeup' from previous mates (e.g. Pitcher et al., 2003).

There were also a range of moderators which did not significantly influence the degree of copying. For example, copying was performed to a similar extent in animals born in the wild or in captivity. This firstly confirms that mate-choice copying is not an artefact of captivity – it is a real behaviour which exists in wild populations. However, the recent meta-analysis by Jones & DuVal (2019) found that females copied more when tested in the wild compared to when tested in captivity. This discrepancy likely arises because our moderator focuses on observer history prior to, but not during, the behavioural tests, and suggests that the environment during the test is more important for influencing copying behaviour. Copying

was not influenced by the mating rate of the species tested. This null result could be explained partly by the lack of data on species with low rates of multiple mating, and we suggest more studies should be carried out on such species. Importantly, any conclusions relating to this moderator may also be further limited by the fact that species with low rates of multiple mating are taxonomically limited in our sample, occurring only in birds and arthropods. Nevertheless, mate-choice copying has been shown in at least two bird species that show some form of social monogamy and biparental care (zebra finches and Japanese quail), even though theory suggests that copying should not be favoured because of the diminishing returns associated with sharing a mate that provides direct benefits or parental care (Vakirtzis and Roberts, 2009). We also found no difference in the strength of generalised versus individual copying. These two forms of copying may require different cognitive abilities, but whether this leads to differences in the strength of copying is not clear. Crucially, the relative prevalence, and strength, of these two forms of copying has important evolutionary consequences. Both forms of copying can widen the gap between attractive and unattractive males (Leadbeater, 2009), create frequency-dependent bias in mate choice (Santos et al., 2014) and support the invasion of new traits into a population (Santos et al., 2017). However, because generalised copying allows learned preferences to be applied to multiple mates, it can lead to the cultural transmission of preferences, and so has the potential to be a much stronger evolutionary force than individual copying.

Most surprisingly, we found no evidence that males copy less than females. This result is unexpected as males may be substantially increasing their exposure to sperm competition by copying the mate choices of their rivals (Simmons, 2001; Vakirtzis and Roberts, 2009). In support of this, there is widespread evidence that males often prefer to mate with virgin females if given the choice (Simmons, 2001; Bonduriansky, 2001). Again, this could be partly explained by the relatively small number of studies testing male copying. Another possibility is that when faced with previously-mated females, males may ejaculate more sperm as a defence against combat sperm competition, instead of rejecting a mating opportunity (Simmons, 2001; Kelly & Jennions, 2011). The ability of males to strategically allocate sperm in this way has the potential to reduce the costs of copying a rival's choice. Copying could also be beneficial to males if they can reduce the risk of being copied themselves, perhaps by mating or courting out of view of rival males (Simmons, 2001; Brown et al., 2012; le Roux et al., 2013), or by mating with less-preferred females when rivals are present as a form of deception (also known as the 'audience effect': Plath et al., 2008; Castellano et al., 2016; Witte et al., 2018). Alternatively, male mate-choice copying may be likely when males face high costs of reproduction, as appears to be the case for the fantail darter Etheostoma flabellare, the three-spined stickleback Gasterosteus aculeatus, and the broadnosed pipefish Syngnathus typhle (Witte et al., 2015). Indeed, mate-choice copying studies have often been performed using species with paternal care (fish, zebra finches), and therefore high male reproductive investment. It will therefore be useful to test for copying in species which exhibit a broader range of mating systems and reproductive strategies than that shown by the current available sample.

Several other aspects of experimental design did not influence the degree of mate choice copying. For example, copying was observed regardless of the type of social information available to the observer (whether the demonstrator mated, courted, avoided or affiliated with a partner), how the observer preference was measured, or whether the demonstrator was allowed to 'choose' a target individual or was artificially placed with one. Nevertheless, variation in other aspects of experimental design do have the potential to contribute to some of the unexplained variability in effect sizes seen across studies. For example, for indirect

measures of mate choice (Rosenthal, 2017), the criteria used to determine mating preferences differs between studies. Vukomanovic and Rodd (2007) conducted mate-choice copying experiments with the sailfin molly *Poecilia latipinna* using different criteria to investigate how this could influence the measurement of copying. Using the criteria of Dugatkin (1992), where females had to spend over 50% of their time within a 13cm "preference zone" in front of the male, significant copying was detected. However, using the criteria of Lafleur et al. (1997), where females had to spend 15 consecutive seconds in a 2.5cm preference zone, no copying was detected. This demonstrates that even when recording the same behaviours, results can differ depending on exactly which methodology is used.

Though several moderators did influence the strength of copying in our sample, most of the heterogeneity in effect size detected (both here and in Jones & DuVal, 2019) remains unexplained. This suggests there are other factors which influence the degree of copying which we were unable to include in our meta-analysis. In many cases this is because of the difficulty associated with obtaining good estimates across all species sampled. For example, a potentially important driver of copying, and mate choice in general, is the cost of mate sampling. However we have very few estimates of the costs of mate sampling and mate choice in any species, so that broad-scale comparisons are difficult (Rosenthal, 2017). Interestingly, empirical studies that have attempted to indirectly measure this effect, by testing copying in high-cost environments, have failed to find any significant change in copying (Briggs et al., 1996; Dugatkin & Godin, 1998). Another key driver of copying is expected to be the ability to assess mate quality, which again will be difficult to accurately estimate for all species. However, empirical tests have found some support for this effect: for example, copying is more likely when mates are more similar in guppies and mollies (Dugatkin, 1996; Witte & Ryan, 1998). Related to this, a valuable topic for future research

would be the nature of memory in the context of mate-choice copying. How long are demonstrations remembered for? How easily are these memories forgotten or overwritten with conflicting experiences?

In summary, this meta-analysis of 58 experimental studies suggests that mate-choice copying is a widespread and robust phenomenon across the animal kingdom. Despite different methodological approaches, the average effect size is very similar to that found in the recent meta-analysis of mate-choice copying by Jones & DuVal (2019), of which our dataset overlaps. We found evidence that the amount of copying differs significantly across taxonomic groups and between the two main types of experimental design found in the literature, but no evidence that copying is influenced by sex or the rate of multiple mating. Importantly, our systematic review has also revealed areas where more empirical work is needed. Unfortunately, though mate-choice copying has been observed across a wide taxonomic range most studies are still only performed on a few model species. In addition to uncovering the details of how mate-choice copying operates within a species, time should be taken to investigate how widespread the phenomenon is across the animal kingdom, and which aspects of a species' ecology or biology predispose the evolution of copying.

References

Studies used in the meta-analysis are indicated with a \star

 * Applebaum SL, Cruz A, 2000. The role of mate-choice copying and disruption effects in mate preference determination of *Limia perugiae* (Cyprinodontiformes, Poeciliidae). Ethology 106:933-944.

- *Auld HL, Godin JGJ, 2015. Sexual voyeurs and copiers: social copying and the audience effect on male mate choice in the guppy. Behavioral Ecology and Sociobiology 69:1795-1807.
- *Auld HL, Punzalan D, Godin JGJ, Rundle HD, 2009. Do female fruit flies (*Drosophila serrata*) copy the mate choice of others? Behavioural Processes 82:78-80.
- Bell AM, Hankison SJ, Laskowski KL, 2009. The repeatability of behaviour: a meta-analysis. Animal Behaviour 77:771-783.
- Blomberg SP, Garland T, Ives AR, 2003. Testing for phylogenetic signal in comparative data: behavioral traits are more labile. Evolution 57:717-745.
- Bonduriansky R, 2001. The evolution of male mate choice in insects: a synthesis of ideas and evidence. Biological Reviews 76:305-339.
- Borenstein M, Hedges LV, Higgins JP, Rothstein H, 2009. Introduction to meta-analysis. Chichester, UK: John Wiley.
- ★Briggs SE, Godin JGJ, Dugatkin LA, 1996. Mate-choice copying under predation risk in the Trinidadian guppy (*Poecilia reticulata*). Behavioral Ecology 7:151-157.
- ★Brooks R, 1996. Copying and the repeatability of mate choice. Behavioral Ecology and Sociobiology 39:323-329.
- ★Brooks R, 1999. Mate choice copying in guppies: Females avoid the place where they saw courtship. Behaviour 136:411-421.
- Brown C, Garwood MP, Williamson JE, 2012. It pays to cheat: tactical deception in a cephalopod social signalling system. Biology letters 8:729-732.
- Castellano S, Friard O, Pilastro A, 2016. The audience effect and the role of deception in the expression of male mating preferences. Animal Behaviour 115:273-282.
- Clutton-Brock T, McComb K, 1993. Experimental tests of copying and mate choice in fallow deer (*Dama dama*). Behavioral Ecology 4:191-193.

Cohen J, 1992. A Power Primer. Psychological Bulletin 112:155-159.

- Cotton S, Small J, Pomiankowski A, 2006. Sexual selection and condition-dependent mate preferences. Current Biology 16:R755-R765.
- *Dagaeff AC, Pocheville A, Nöbel S, Loyau A, Isabel G, Danchin E, 2016. Drosophila mate copying correlates with atmospheric pressure in a speed learning situation. Animal Behaviour 121:163-174.
- * Danchin E, Nöbel S, Pocheville A, Dagaeff AC, Demay L, Alphand M, Ranty-Roby S, Van Renssen L, Monier M, Gazagne E, Allain M, Isabel G, 2018. Cultural flies: Conformist social learning in fruitflies predicts long-lasting mate-choice traditions. Science 362:1025-1030.
- *Doucet SM, Yezerinac SM, Montgomerie R, 2004. Do female zebra finches (*Taeniopygia guttata*) copy each other's mate preferences? Canadian Journal of Zoology 82:1-7.
- Dougherty LR, 2020. Designing mate choice experiments. Biological Reviews 95: 759-781.
- Dougherty LR, Shuker DM, 2015. The effect of experimental design on the measurement of mate choice: a meta-analysis. Behavioral Ecology 26:311-319.
- Drullion D, Dubois F, 2008. Mate-choice copying by female zebra finches, *Taeniopygia guttata*: what happens when model females provide inconsistent information?
 Behavioral Ecology and Sociobiology 63:269-276.
- Dubois F, Drullion D, Witte K, 2012. Social information use may lead to maladaptive decisions: a game theoretic model. Behavioral Ecology 23:225-231.
- Dugatkin LA, 1992. Sexual selection and imitation: females copy the mate choice of others. American Naturalist 139:1384-1389.
- Dugatkin LA, 1996. Interface between culturally based preferences and genetic preferences: female mate choice in Poecilia reticulata. Proceedings of the National Academy of Sciences 93:2770-2773.

- ★Dugatkin LA, 1998. Genes, copying, and female mate choice: Shifting thresholds. Behavioral Ecology 9:323-327.
- ★Dugatkin LA, Druen MW, Godin JGJ, 2003. The disruption hypothesis does not explain mate-choice copying in the guppy (*Poecilia reticulata*). Ethology 109:67-76.
- *Dugatkin LA, Godin JGJ, 1992. Reversal of female mate choice by copying in the guppy (*Poecilia reticulata*). Proceedings of the Royal Society B: Biological Sciences 249:179-184.
- Dugatkin LA, Godin JGJ, 1993. Female mate copying in the guppy (*Poecilia reticulata*):
 Age-dependent effects. Behavioral Ecology 4:289-292.
- ★Dugatkin LA, Godin JGJ, 1998. Effects of hunger on mate-choice copying in the guppy. Ethology 104:194-202.
- Duval S, Tweedie R, 2000. Trim and fill: a simple funnel-plot–based method of testing and adjusting for publication bias in meta-analysis. Biometrics 56:455-463.
- Egger M, Smith GD, Schneider M, Minder C, 1997. Bias in meta-analysis detected by a simple, graphical test. BMJ 315:629-634.
- *Forsgren E, Karlsson A, Kvarnemo C, 1996. Female sand gobies gain direct benefits by choosing males with eggs in their nests. Behavioral Ecology and Sociobiology 39:91-96.
- ★Fowler-Finn KD, Sullivan-Beckers L, Runck AM, Hebets EA, 2015. The complexities of female mate choice and male polymorphisms: Elucidating the role of genetics, age, and mate-choice copying. Current Zoology 61:1015-1035.
- *Frommen JG, Rahn AK, Schroth SH, Waltschyk N, Bakker TCM, 2009. Mate-choice copying when both sexes face high costs of reproduction. Evolutionary Ecology 23:435-446.

- ★Galef BG, Lim TCW, Gilbert GS, 2008. Evidence of mate choice copying in Norway rats, *Rattus norvegicus*. Animal Behaviour 75:1117-1123.
- *Galef BG, White DJ, 1998. Mate-choice copying in Japanese quail, *Coturnix coturnix japonica*. Animal Behaviour 55:545-552.
- Germain M, Blanchet S, Loyau A, Danchin E, 2016. Mate-choice copying in *Drosophila melanogaster*: Impact of demonstration conditions and male-male competition.
 Behavioural Processes 125:76-84.
- ★Gierszewski S, Keil M, Witte K, 2018. Mate-choice copying in sailfin molly females: public information use from long-distance interactions. Behavioral Ecology and Sociobiology 72.
- Gildersleeve K, Haselton MG, Fales MR, 2014. Do women's mate preferences change across the ovulatory cycle? A meta-analytic review. Psychological Bulletin 140:1205.
- Godin J-GJ, Herdman EJ, Dugatkin LA, 2005. Social influences on female mate choice in the guppy, Poecilia reticulata: generalized and repeatable trait-copying behaviour. Animal Behaviour 69:999-1005.
- ★Godin JGJ, Hair KPE, 2009. Mate-choice copying in free-ranging Trinidadian guppies (*Poecilia reticulata*). Behaviour 146:1443-1461.
- ★Goldschmidt T, Bakker TCM, Feuth-De Bruijn E, 1993. Selective copying in mate choice of female sticklebacks. Animal Behaviour 45:541-547.
- Gouda-Vossos A, Nakagawa S, Dixson BJW, C. BR, 2018. Mate choice copying in humans: a systematic review and meta-analysis. Adaptive Human Behavior and Physiology 4:364-386.
- Grafen A, 1989. The phylogenetic regression. Philosophical Transactions of the Royal Society B 326:119-157.

- ★ Grant JWA, Green LD, 1996. Mate copying versus preference for actively courting males by female Japanese medaka (*Oryzias latipes*). Behavioral Ecology 7:165-167.
- Hadfield JD, 2010. MCMC Methods for Multi-Response Generalized Linear Mixed Models: The MCMCglmm R Package. Journal of Statistical Software 33:1-22.
- Harts AM, Booksmythe I, Jennions MD, 2016. Mate guarding and frequent copulation in birds: A meta-analysis of their relationship to paternity and male phenotype. Evolution 70:2789-2808.
- Hedges LV, Olkin I, 1985. Statistical methods for meta-analysis. Orlando, FL: Academic Press.
- *Heubel KU, Hornhardt K, Ollmann T, Parzefall J, Ryan MJ, Schlupp I, 2008. Geographic variation in female mate-copying in the species complex of a unisexual fish, *Poecilia formosa*. Behaviour 145:1041-1064.
- Higgins J, Thompson SG, Deeks JJ, Altman DG, 2003. Measuring inconsistency in metaanalyses. British Medical Journal 327:557-560.
- ★Hill SE, Ryan MJ, 2006. The role of model female quality in the mate choice copying behaviour of sailfin mollies. Biology Letters 2:203-205.
- *Howard RD, Martens RS, Innis SA, Drnevich JM, Hale J, 1998. Mate choice and mate competition influence male body size in Japanese medaka. Animal Behaviour 55:1151-1163.
- Jennions MD, Møller AP, 2002. Relationships fade with time: a meta-analysis of temporal trends in publication in ecology and evolution. Proceedings of the Royal Society B: Biological Sciences 269:43-48.
- Jennions MD, Petrie M, 1997. Variation in mate choice and mating preferences: a review of causes and consequences. Biological Reviews 72:283-327.

- Jones BC, DuVal EH, 2019. Mechanisms of social influence: a meta-analysis of the effects of social information on female mate choice decisions. Frontiers in Ecology and Evolution 7:390.
- Kavaliers M, Choleris E, Agmo A, Braun WJ, Colwell DD, Muglia LJ, Ogawa S, Pfaff DW, 2006. Inadvertent social information and the avoidance of parasitized male mice: A role for oxytocin. Proceedings of the National Academy of Sciences of the United States of America 103:4293-4298.
- Kelly CD, Jennions MD, 2011. Sexual selection and sperm quantity: meta-analyses of strategic ejaculation. Biological Reviews 86:863-884.
- Kniel N, Schmitz J, Witte K, 2015. Quality of public information matters in mate-choice copying in female zebra finches. Frontiers in Zoology 12.
- Kokko H, Brooks R, Jennions MD, Morley J, 2003. The evolution of mate choice and mating biases. Proceedings of the Royal Society B 270:653-664.
- Koricheva J, Gurevitch J, Mengeresen K, 2013. Handbook of meta-analysis in ecology and evolution. Princeton: Princeton University Press.
- *Lafleur DL, Lozano GA, Sclafani M, 1997. Female mate-choice copying in guppies, *Poecilia reticulata*: A re-evaluation. Animal Behaviour 54:579-586.
- Lakens D, Hilgard J, Staaks J, 2016. On the reproducibility of meta-analyses: Six practical recommendations. BMC Psychology 4:24.
- Le Roux A, Snyder-Mackler N, Roberts EK, Beehner JC, Bergman TJ, 2013. Evidence for tactical concealment in a wild primate. Nature Communications 4:1-6.

Leadbeater E, 2009. Social learning: what do Drosophila have to offer? Current Biology 19:R378-R380.

- *Loyau A, Blanchet S, Van Laere P, Clobert J, Danchin E, 2012. When not to copy: Female fruit flies use sophisticated public information to avoid mated males. Scientific Reports 2. doi: 10.1038/srep00768.
- * Matsumoto Y, Takegaki T, 2013. Female mate choice copying increases egg survival rate but does not reduce mate-sampling cost in the barred-chin blenny. Animal Behaviour 86:339-346.
- *McComb K, Clutton-Brock T, 1994. Is mate choice copying or aggregation responsible for skewed distributions of females on leks? Proceedings of the Royal Society B: Biological Sciences 255:13-19.
- *Mery F, Varela SAM, Danchin E, Blanchet S, Parejo D, Coolen I, Wagner RH, 2009. Public Versus Personal Information for Mate Copying in an Invertebrate. Current Biology 19:730-734.
- Michonneau F, Brown JW, Winter DJ, 2016. rotl: and R package to interact with the Open Tree of Life data. Methods in Ecology and Evolution 7:1476-1481.
- *Milner RNC, Jennions MD, Backwell PRY, 2011. Non-independent mate choice in a fiddler crab: A case of stimulus enhancement. Behavioral Ecology and Sociobiology 65:1419-1424.
- *Moran RL, von Ende CN, King BH, 2013. Mate choice copying in two species of darters (Percidae: Etheostoma). Behaviour 150:1255-1274.
- *Munger L, Cruz A, Applebaum S, 2004. Mate choice copying in female humpback limia (Limia nigrofasciata, family Poeciliidae). Ethology 110:563-573.
- Nakagawa S, Santos ES, 2012. Methodological issues and advances in biological metaanalysis. Evolutionary Ecology 26:1253-1274.
- Nichols CR, 1991. A comparison of the reproductive and behavioural differences in feral and domestic Japanese quail. PhD thesis: University of British Columbia.

- ★Nöbel S, Allain M, Isabel G, Danchin E, 2018a. Mate copying in *Drosophila melanogaster* males. Animal Behaviour 141:9-15.
- *Nöbel S, Danchin E, Isabel G, 2018b. Mate-copying for a costly variant in *Drosophila melanogaster* females. Behavioral Ecology 29:1150-1156.
- Norton S, Uetz GW, 2005. Mating frequency in *Schizocosa ocreata* (Hentz) wolf spiders: evidence for a mating system with female monandry and male polygyny. The Journal of Arachnology 33:16-25.
- ★ Ophir AG, Galef BG, 2003. Female Japanese quail affiliate with live males that they have seen mate on video. Animal Behaviour 66:369-375.
- ★ Ophir AG, Galef BG, 2004. Sexual experience can affect use of public information in mate choice. Animal Behaviour 68:1221-1227.
- Paradis E, Claude J, Strimmer K, 2004. APE: analyses of phylogenetics and evolution in R language. Bioinformatics 20:289-290.
- *Patriquin-Meldrum KJ, Godin JGJ, 1998. Do female three-spined sticklebacks copy the mate choice of others? American Naturalist 151:570-577.
- Pitcher TE, Neff BD, Rodd FH, Rowe L, 2003. Multiple mating and sequential mate choice in guppies: females trade up. Proceedings of the Royal Society of London Series B: Biological Sciences 270:1623-1629.
- Plath M, Richter S, Tiedemann R, Schlupp I, 2008. Male fish deceive competitors about mating preferences. Current Biology 18:1138-1141.
- Pruett-Jones S, 1992. Independent versus nonindependent mate choice: do females copy each other? The American Naturalist 140:1000-1009.
- Ptacek M, Breden F, 1998. Phylogenetic relationships among the mollies (Poeciliidae:Poecilia: Mollienesia group) based on mitochondrial DNA sequences. Journal of FishBiology 53:64-81.

- R Development Core Team, 2018. R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing.
- Reznick DN, Furness AI, Meredith RW, Springer MS, 2017. The origin and biogeographic diversification of fishes in the family Poeciliidae. PloS one 12:e0172546.

Rohatgi A, 2019. WebPlotDigitizer. 4rd edn.

Rosenberg MS, 2005. The file-drawer problem revisited: a general weighted method for calculating fail-safe numbers in meta-analysis. Evolution 59:464-468.

Rosenthal GG, 2017. Mate choice. Princeton: Princeton University Press.

- Santos M, Matos M, Varela SA, 2014. Negative public information in mate choice copying helps the spread of a novel trait. The American Naturalist 184:658-672.
- Santos M, Sapage M, Matos M, Varela SA, 2017. Mate-choice copying: A fitness-enhancing behavior that evolves by indirect selection. Evolution 71:1456-1464.
- Scauzillo RC, Ferkin MH, 2019. Factors that affect non-independent mate choice. Biological Journal of the Linnean Society 128:499-514.
- Schartl M, Wilde B, Schlupp I, Parzefall J, 1995. Evolutionary origin of a parthenoform, the Amazon molly Poecilia formosa, on the basis of a molecular genealogy. Evolution 49:827-835.
- *Schlupp I, Marler C, Ryan MJ, 1994. Benefit to male sailfin mollies of mating with heterospecific females. Science 263:373-374.
- *Schlupp I, Ryan MJ, 1997. Male sailfin mollies (Poecilia latipinna) copy the mate choice of other males. Behavioral Ecology 8:104-107.
- Shettleworth SJ, 2001. Animal cognition and animal behaviour. Animal Behaviour 61:277-286.
- Silberzahn R, Uhlmann EL, Martin DP, Anselmi P, Aust F, Awtrey E, Bahník Š, Bai F, Bannard C, Bonnier E, 2018. Many analysts, one data set: Making transparent how

variations in analytic choices affect results. Advances in Methods and Practices in Psychological Science 1:337-356.

- Simmons LW, 2001. Sperm competition and its evolutionary consequences in the insects. Princeton: Princeton University Press.
- *Slagsvold T, Viljugrein H, 1999. Mate choice copying versus preference for actively displaying males by female pied flycatchers. Animal Behaviour 57:679-686.
- Swaddle JP, Cathey MG, Correll M, Hodkinson BP, 2005. Socially transmitted mate preferences in a monogamous bird: A non-genetic mechanism of sexual selection.
 Proceedings of the Royal Society B: Biological Sciences 272:1053-1058.
- *Swaddle JP, Clelland E, 2009. Deciding with whom to Mate: Do Female Finches Follow Fashion? In: Murphy D, Longo D, editors. Encyclopedia of Psychology of Decision Making: Nova Science Publishers, Inc. p. 663-676.
- Trivers RL, 1972. Parental investment and sexual selection. In: Campbell B, editor. Sexual Selection and the Descent of Man, 1871-1971 Chicago: Aldine. p. 795-838.
- Vakirtzis A, 2011. Mate choice copying and nonindependent mate choice: A critical review. Annales Zoologici Fennici 48:91-107.
- Vakirtzis A, Roberts SC, 2009. Mate choice copying and mate quality bias: Different processes, different species. Behavioral Ecology 20:908-911.
- Varela SA, Matos M, Schlupp I, 2018. The role of mate-choice copying in speciation and hybridization. Biological Reviews 93:1304-1322.
- Viechtbauer W, 2010. Conducting meta-analyses in R with the metafor package. Journal of Statistical Software 36:1-48.
- Vukomanovic J, Rodd FH, 2007. Size-dependent female mate copying in the guppy (*Poecilia reticulata*): Large females are role models but small ones are not. Ethology 113:579-586.

- Wade MJ, Pruett-Jones SG, 1990. Female copying increases the variance in male mating success. Proceedings of the National Academy of Sciences 87:5749-5753.
- Wedell N, Gage MJG, Parker GA, 2002. Sperm competition, male prudence and spermlimited females. Trends in Ecology & amp; Evolution 17:313-320
- ★White DJ, Galef BG, 1999. Mate choice copying and conspecific cueing in Japanese quail, Coturnix coturnix japonica. Animal Behaviour 57:465-473.
- ★White DJ, Galef BG, 2000a. 'Culture' in quail: Social influences on mate choices of female Coturnix japonica. Animal Behaviour 59:975-979.
- ★White DJ, Galef BG, 2000b. Differences between the sexes in direction and duration of response to seeing a potential sex partner mate with another. Animal Behaviour 59:1235-1240.
- * White DJ, Watts E, Pitchforth K, Agyapong S, Miller N, 2017. 'Sociability' affects the intensity of mate-choice copying in female guppies, *Poecilia reticulata*. Behavioural Processes 141:251-257.
- Wicherts JM, Veldkamp CL, Augusteijn HE, Bakker M, Van Aert R, Van Assen MA, 2016.
 Degrees of freedom in planning, running, analyzing, and reporting psychological studies: A checklist to avoid p-hacking. Frontiers in psychology 7:1832.
- ★Widemo MS, 2006. Male but not female pipefish copy mate choice. Behavioral Ecology 17:255-259.
- Witte K, Baumgärtner K, Röhrig C, Nöbel S, 2018. Test of the deception hypothesis in Atlantic mollies Poecilia mexicana—Does the audience copy a pretended mate choice of others? Biology 7:40.
- Witte K, Kniel N, Kureck IM, 2015. Mate-choice copying: status quo and where to go. Current Zoology 61:1073-1081.

- ★Witte K, Massmann R, 2003. Female sailfin mollies, *Poecilia latipinna*, remember males and copy the choice of others after 1 day. Animal Behaviour 65:1151-1159.
- *Witte K, Noltemeier B, 2002. The role of information in mate-choice copying in female sailfin mollies (*Poecilia latipinna*). Behavioral Ecology and Sociobiology 52:194-202.
- *Witte K, Ryan MJ, 1998. Male body length influences mate-choice copying in the sailfin molly *Poecilia latipinna*. Behavioral Ecology 9:534-539.
- *Witte K, Ueding K, 2003. Sailfin molly females (*Poecilia latipinna*) copy the rejection of a male. Behavioral Ecology 14:389-395
- ★Wong BBM, McCarthy M, 2009. Prudent male mate choice under perceived sperm competition risk in the eastern mosquito fish. Behavioral Ecology 20:278-282.
- Wood W, Kressel L, Joshi PD, Louie B, 2014. Meta-analysis of menstrual cycle effects on women's mate preferences. Emotion Review 6:229-249.

Table 1. Results of meta-regression models testing the effecting of ten moderators on the strength of copying. The effect of each factor was tested using a mixed-effects model, with study ID, species, phylogeny and observation ID as random factors, and one of the ten moderators as a fixed factor. Significance was determined using the Q_M statistic. All moderators were tested using the full dataset (158 effect sizes), with the exception of the rate of multiple mating, which was tested using a reduced dataset (149 effect sizes). Significant moderators are highlighted in bold.

Moderator	<i>Q</i> _M	Р
Taxonomic group	8.57	0.04
Choosing sex	0.05	0.82
Rate of multiple mating	0.15	0.70
Animal birth location	1.25	0.53
Generalised or individual copying	0.57	0.45
Design type	6.11	0.01
Model pairing	0.01	0.92
Demonstrator behaviour	2.00	0.57
Preference measure	6.37	0.27
Study year	1.97	0.16



Figure 1. PRISMA diagram showing the literature search process.



Figure 2. The full phylogenetic tree used in the meta-analysis, with taxonomic groups highlighted. Note that the arthropod category includes arachnid, crustacean and insect species.



Figure 3. The number of effect sizes in the analysis in relation to: a) the four taxonomic groups, and b) the two sexes. In panel a), the different colours show the number of effect sizes for the three most commonly-studies species (*Poecilia reticulata*, *Drosophila melanogaster*, and *Poecilia latipinna*), which together account for 61% of all effect sizes in the analysis.



Figure 4: Funnel plot showing the relationship between inverse standard error (precision) and effect size (the standardised mean difference, Hedges'*d*). The dashed vertical line indicates the mean effect size from a random-effects-only model (0.57).



Figure 5. Forest plot showing the mean effect size (the standardised mean difference, Hedges' d) for each factor level for two moderators: taxonomic class (yellow diamonds) and design type (blue diamonds). The mean effect size for the whole dataset is shown in black for comparison. See text for details on the two moderators. Bars indicate 95% confidence intervals. k= the number of effect sizes for each factor level.





Supplementary methods and results for:

Davies, AD¹, Lewis, Z² & Dougherty, LR^{1*} 2020. A meta-analysis of factors influencing the strength of mate choice copying in animals. Behavioral Ecology.

¹Institute of Integrative Biology, University of Liverpool, Liverpool, L69 7ZB, UK ²School of Life Sciences, University of Liverpool, Liverpool, L69 7ZB, UK *Corresponding author email: liam.dougherty@liverpool.ac.uk **Table S1:** Resons for rejecting papers that were read in full.

Study	Species	Explanation
Alonzo, 2008	Symphodus ocellatus	Missing sample sizes
Amlacher and Dugatkin, 2005	Poecilia reticulata	No control data
Bailey and Zuk, 2009	Teleogryllus oceanicus	Not mate choice copying
Bierbach <i>et al.,</i> 2013a	Poecilia mexicana	Statistics not appropriate
Bierbach <i>et al.,</i> 2011	Poecilia mexicana	Statistics not appropriate
Bierbach <i>et al.,</i> 2013b	Several	Not mate choice copying
Castellano <i>et al.,</i> 2016	N/A	Model: no data
Dugatkin, 1992	Poecilia reticulata	Statistics not appropriate
Dugatkin, 1996	Poecilia reticulata	Statistics not appropriate
Dugatkin, 2007	Poecilia reticulata	Statistics not appropriate
Dugatkin and Druen, 2007	Poecilia reticulata	Statistics not appropriate
Dugatkin <i>et al.,</i> 2002	Poecilia reticulata	Not mate choice copying
Fiske <i>et al.,</i> 1996	Gallinago media	Observational study
Freed-Brown and White, 2009	Molothrus ater	Missing sample sizes
Godin <i>et al.,</i> 2005	Poecilia reticulata	Statistics not appropriate
Goncalves <i>et al.,</i> 2003	Salaria pavo	Not mate choice copying
Coulot and Coulot 2000	Amblyglyphidodon	Observational study
Goulet and Goulet, 2006	leucogaster	Observational study
Höglund <i>et al.,</i> 1995	Tatrao tetrix	Missing sample sizes
Kavaliers <i>et al.,</i> 2006	Mus musculus	Indirect social information

Kniel <i>et al.,</i> 2015a	Taeniopygia guttata
Kniel <i>et al.,</i> 2015b	Taeniopygia guttata
Kniel <i>et al.,</i> 2017	Taeniopygia guttata
Makowicz et al., 2010	Poecilia reticulata
Monier <i>et al.,</i> 2018	Drosophila melanogaster
Monier <i>et al.,</i> 2019	Drosophila melanogaster
Ohlyan <i>et al.,</i> 2012	Poecilia reticulata
Persaud and Galef, 2005	Coturnix japonica
Reynolds and Jones, 1999	Pomatoschistus microps
Sakaluk and Ivy, 1999	Cyphoderris strepitans
Witte and Ryan, 2002	Poecilia latipinna
Witte <i>et al.,</i> 2018	Poecilia mexicana

Statistics not appropriate Statistics not appropriate Statistics not appropriate No control data Statistics not appropriate Not mate choice copying Missing sample sizes Not mate choice copying Not mate choice copying Missing sample sizes Statistics not appropriate **Table S2**. Comparison of key methodological differences between the current study and the

recent meta-analysis of mate-choice copying by Jones & DuVal (2019: see text for

reference).

Stage	Methodology	Jones & DuVal, 2019	Davies et al, 2020
Literature	Search date	May 2019	April 2019
search	Sources	Web of Science	Web of Science, Scopus,
			Studies cited by and citing
			Vakirtzis (2011)
	Articles screened	365	844
	Number of assessors	1	1
Study	Humans	No	No
selection	Sex	Females	Females and males
	Control group	Not required	Not required
	Copying type	Individual and generalised	Individual and generalised
	Articles assessed	91	89
Data	Effect size	Odds ratio	Standardised mean
extraction			difference (Hedges' d)
	Data type	Dichotomous	Dichotomous, continuous
	Re-analyses?	No	Yes
	Contacting authors?	No	Yes
	Directionless effect	Excluded	Included
	sizes?		
Sample	Included studies	40	58
sizes	Included species	17	23
	Included effect sizes	103	158
Analyses	Approach	Bayesian	Restricted maximum
			likelihood
	Package	MCMCglmm	Metafor
	Random effects	Study ID, species,	Study ID, species,
		phylogeny, observation	phylogeny, observation
	Heterogeneity	²	12
	Meta-regressions	Single model (4	Separate models (9
		moderators)	moderators)
Publication	Temporal trends	None	Meta-regression
bias	Asymmetry	Modified trim-and-fill	Modified trim-and-fill
		(residuals)	(residuals)
		Modified Egger's	Modified Egger's

Table S3: Sources used to classify species according to their rate of multiple mating within a breeding season. Where possible we used

paternity data or quantitative measurements of mating rates.

Taxonomic group	Latin name	Multiple mating rate	Source	Evidence
Arthropod	Drosophila melanogaster	High	Imhof et al., 2002	Paternity data
	Drosophila serrata	High	Frentiu et al., 2008	Paternity data
	Schizocosa sp.	Low	Norton & Uetz, 2005	Mating observations
	Uca mjoebergi	Low	Vega-Trejo & Backwell, 2017	Descriptive: "Wandering females approach and briefly enter the burrows of one or more courting males before choosing a mate by remaining in a breeding burrow" pp 170
Fish	Etheostoma flabellare	High	Moran et al., 2013	Descriptive: "multiple females approach and lay eggs in the nest." pp 1258. "The females then attach their eggs in the nests of multiple males." pp 1268
	Etheostoma zonale	High	Moran et al., 2013	Descriptive: "females approach males and attach eggs to nearby algae." pp 1258
	Gambusia holbrooki	High	Zane et al., 1999	Paternity data
	Gasterosteus aculeatus	High	Frommen et al., 2009	Spawning observations
	Limia nigrofasciata	Unknown	NA	ΝΑ
	Limia perugiae	High	Schartl et al., 1993	Paternity data
	Oryzias latipes	High	Weir & Grant, 2010	Spawning observations

Poecilia formosaN/AHubbs & Hubbs, 1932ParthenogeneticPoecilia latipinnaHighTravis et al., 1990Paternity dataPoecilia mexicanaUnknownNANAPoecilia reticulataHighHoude, 1997Paternity dataPomatoschistus minutusHighForsgren et al., 2006Spawning observationsRhabdoblennius nitidusHighJones et al., 1999Paternity dataBirdCoturnix japonicaLowNichols, 1991Paternity dataFicedula hypoleucaLowBirkhead et al., 1990Paternity dataMammalDama damaHighApollonio et al., 1992Paternity dataMammalDama damaHighApollonio et al., 1992Paternity data		i de la companya de l		i de la constante de la constan	
Poecilia latipinnaHighTravis et al., 1990Paternity dataPoecilia mexicanaUnknownNANAPoecilia reticulataHighHoude, 1997Paternity dataPomatoschistus minutusHighForsgren et al., 2006Spawning observationsRhabdoblennius nitidusHighMiyano et al., 2006Spawning observationsBirdCoturnix japonicaLowNichols, 1991Paternity dataFicedula hypoleucaLowNichols, 1991Paternity dataMammalDama damaHighApollonio et al., 1990Paternity dataMammalDama damaHighApollonio et al., 1990Paternity data		Poecilia formosa	N/A	Hubbs & Hubbs, 1932	Parthenogenetic
Poecilia mexicanaUnknownNAPoecilia reticulataHighHoude, 1997Paternity dataPomatoschistus minutusHighForsgren et al., 2006Spawning observationsRhabdoblennius nitidusHighMiyano et al., 2006Spawning observationsBirdCoturnix japonicaLowNichols, 1999Paternity dataFicedula hypoleucaLowBirkhead et al., 1999Paternity dataMammalDama damaHighApollonio et al., 1991Paternity dataMammalDama damaHighBirkhead et al., 1990Paternity dataMammalDama damaHighApollonio et al., 2016Paternity data		Poecilia latipinna	High	Travis et al., 1990	Paternity data
Poecilia reticulataHighHoude, 1997Paternity dataPomatoschistus minutusHighForsgren et al., 2006Spawning observationsRhabdoblennius nitidusHighMiyano et al., 2006Spawning observationsSyngnathus typhleHighJones et al., 1999Paternity dataBirdCoturnix japonicaLowNichols, 1991Paternity dataFicedula hypoleucaLowBirkhead et al., 1990Paternity dataMammalDama damaHighApollonio et al., 1992Paternity dataMatus norvegicusHighGlass et al., 2016Paternity data		Poecilia mexicana	Unknown	NA	NA
Pomatoschistus minutusHighForsgren et al., 2006Spawning observationsRhabdoblennius nitidusHighMiyano et al., 2006Spawning observationsSyngnathus typhleHighJones et al., 1999Paternity dataBirdCoturnix japonicaLowNichols, 1991Pairing observationsFicedula hypoleucaLowBirkhead et al., 1990Paternity dataMammalDama damaHighApollonio et al., 1992Mating observationsRattus norvegicusHighGlass et al., 2016Paternity data		Poecilia reticulata	High	Houde, 1997	Paternity data
Rhabdoblennius nitidusHighMiyano et al., 2006Spawning observationsSyngnathus typhleHighJones et al., 1999Paternity dataBirdCoturnix japonicaLowNichols, 1991Pairing observationsFicedula hypoleucaLowLundberg & Alatalo, 1992Paternity dataMammalDama damaHighApollonio et al., 1992Mating observationsRattus norvegicusHighGlass et al., 2016Paternity data		Pomatoschistus minutus	High	Forsgren et al., 2006	Spawning observations
BirdSyngnathus typhleHighJones et al., 1999Paternity dataBirdCoturnix japonicaLowNichols, 1991Pairing observationsFicedula hypoleucaLowLundberg & Alatalo, 1992Paternity dataTaeniopygia guttataLowBirkhead et al., 1990Paternity dataMammalDama damaHighApollonio et al., 1992Mating observationsRattus norvegicusHighGlass et al., 2016Paternity data		Rhabdoblennius nitidus	High	Miyano et al., 2006	Spawning observations
BirdCoturnix japonicaLowNichols, 1991Pairing observationsFicedula hypoleucaLowLundberg & Alatalo, 1992Paternity dataTaeniopygia guttataLowBirkhead et al., 1990Paternity dataMammalDama damaHighApollonio et al., 1992Mating observationsRattus norvegicusHighGlass et al., 2016Paternity data		Syngnathus typhle	High	Jones et al., 1999	Paternity data
Ficedula hypoleucaLowLundberg & Alatalo, 1992Paternity dataTaeniopygia guttataLowBirkhead et al., 1990Paternity dataMammalDama damaHighApollonio et al., 1992Mating observationsRattus norvegicusHighGlass et al., 2016Paternity data	Bird	Coturnix japonica	Low	Nichols, 1991	Pairing observations
Taeniopygia guttataLowBirkhead et al., 1990Paternity dataMammalDama damaHighApollonio et al., 1992Mating observationsRattus norvegicusHighGlass et al., 2016Paternity data		Ficedula hypoleuca	Low	Lundberg & Alatalo, 1992	Paternity data
MammalDama damaHighApollonio et al., 1992Mating observationsRattus norvegicusHighGlass et al., 2016Paternity data		Taeniopygia guttata	Low	Birkhead et al., 1990	Paternity data
Rattus norvegicus High Glass et al., 2016 Paternity data	Mammal	Dama dama	High	Apollonio et al., 1992	Mating observations
		Rattus norvegicus	High	Glass et al., 2016	Paternity data

References not cited in the text

- Alonzo, S. H. 2008. Female mate choice copying affects sexual selection in wild populations of the ocellated wrasse. Animal Behaviour, 75 (5), 1715-1723.
- Amlacher J, Dugatkin LA, 2005. Preference for older over younger models during matechoice copying in young guppies. Ethology Ecology & Evolution 17:161-169.
- Apollonio M, Festa-Bianchet M, Maro F, Mattioli S, Sarno B, 1992. To let or not to lek: mating strategies of male fallow deer. Behavioral Ecology 3:25-31.
- Bailey NW, Zuk M, 2009. Field crickets change mating preferences using remembered social information. Biology Letters 5:449-451.
- Bierbach D, Jung CT, Hornung S, Streit B, Plath M, 2013a. Homosexual behaviour increases male attractiveness to females. Biology Letters 9. 20121038.
- Bierbach D, Konstantin Y, Stadler S, Schenkel D, Makowicz AM, Nigl H, Geupel H, Riesch R,
 Schlupp I, Plath M, 2013b. Presence of an audience alters mate choice of poeciliid
 males. In: Clark KB, editor. Social Learning Theory: Phylogenetic Considerations
 across Animal, Plant, and Microbial Taxa New York: Nova Science Publishers, Inc. p.
 95-112.
- Bierbach D, Kronmarck C, Hennige-Schulz C, Stadler S, Plath M, 2011. Sperm competition risk affects male mate choice copying. Behavioral Ecology and Sociobiology 65:1699-1707.
- Birkhead TR, Burke T, Zann R, Hunter FM Krupa AP, 1990. Extra-pair paternity and intraspecific brood parasitism in wild zebra finches *Taeniopygia guttata*, revealed by DNA fingerprinting. Behavioral Ecology and Sociobiology 27:315-324.

- Dugatkin LA, 1996. Interface between culturally based preferences and genetic preferences: Female mate choice in *Poecilia reticulata*. Proceedings of the National Academy of Sciences of the United States of America 93:2770-2773.
- Dugatkin, L. A. 2007. Developmental environment, cultural transmission, and mate-choice copying. Naturwissenschaften, 94 (8), 651-656.
- Dugatkin LA, Druen M, 2007. Mother-offspring correlation and mate-choice copying behavior in guppies. Ethology Ecology & Evolution 19:137-144.
- Dugatkin LA, Lucas JS, Godin JGJ, 2002. Serial effects of mate-choice copying in the guppy (*Poecilia reticulata*). Ethology Ecology & Evolution 14:45-52.
- Fiske P, Kålås JA, Sæther SA, 1996. Do female great snipe copy each other's mate choice? Animal Behaviour 51:1355-1362.
- Freed-Brown G, White DJ, 2009. Acoustic mate copying: Female cowbirds attend to other females' vocalizations to modify their song preferences. Proceedings of the Royal Society B: Biological Sciences 276:3319-3325.
- Frentiu FD, Chenoweth SF, 2008. Polyandry and paternity skew in natural and experimental populations of *Drosophila serrata*. Molecular Ecology 17:1589-1596.
- Glass GE, Klein SL, Norris DE, Gardner LC, 2016. Multiple paternity in urban Norway rats: extended ranging for mates. Vector-Borne and Zoonotic Diseases 16:342-348.
- Goncalves D, Oliveira RF, Korner K, Schlupp I, 2003. Intersexual copying by sneaker males of the peacock blenny. Animal Behaviour 65:355-361.
- Goulet D, Goulet TL, 2006. Nonindependent mating in a coral reef damselfish: Evidence of mate choice copying in the wild. Behavioral Ecology 17:998-1003.
- Höglund J, Alatalo RV, Gibson RM, Lundberg A, 1995. Mate-choice copying in black grouse. Animal Behaviour 49:1627-1633.

- Houde AE, 1997. Sex, Color and Mate Choice in Guppies. Princeton: Princeton University Press.
- Hubbs CL, Hubbs LC, 1932. Apparent parthenogenesis in nature in a form of fish of hybrid origin. Science 76:628-630.
- Imhof M, Harr B, Brem G, Schlötterer C, 1998. Multiple mating in wild *Drosophila melanogaster* revisited by microsatellite analysis. Molecular Ecology 7:915-917.
- Kniel N, Dürler C, Hecht I, Heinbach V, Zimmermann L, Witte K, 2015b. Novel mate preference through mate-choice copying in zebra finches: Sexes differ. Behavioral Ecology 26:647-655.
- Kniel N, Müller K, Witte K, 2017. The role of the model in mate-choice copying in female zebra finches. Ethology 123:412-418.

Lundberg A, Alatalo R, 1992. The Pied Flycatcher. London: T A & D poyser.

- Miyano T, Takegaki T, Natsukari Y, 2006. Spawning and egg-tending behavior of the barredchin blenny *Rhabdoblennius ellipes*. Bulletin of the Faculty of Fisheries-Nagasaki University (Japan):1-5.
- Monier M, Nöbel S, Danchin E, Isabel G, 2019. Dopamine and serotonin are both required for mate-copying in Drosophila melanogaster. Frontiers in Behavioral Neuroscience 12. doi: 10.3389/fnbeh.2018.00334.
- Monier M, Nöbel S, Isabel G, Danchin E, 2018. Effects of a sex ratio gradient on female mate-copying and choosiness in Drosophila melanogaster. Current Zoology 64:251-258.
- Ohlyan S, Sihag RC, Yadava NK, 2013. Courtship behaviour and mate choice in guppies: Tactics and strategies. Journal of Nature Science and Sustainable Technology. p. 151-168.

- Persaud KN, Galef BG, 2005. Eggs of a female Japanese quail are more likely to be fertilized by a male that she prefers. Journal of Comparative Psychology 119:251-256.
- Reynolds JD, Jones JC, 1999. Female preference for preferred males is reversed under low oxygen conditions in the common goby (*Pomatoschistus microps*). Behavioral Ecology 10:149-154.
- Sakaluk SK, Ivy TM, 1999. Virgin-male mating advantage in sagebrush crickets: Differential male competitiveness or non-independent female mate choice? Behaviour 136:1335-1346.
- Schartl M, Holter S, Erbelding-Denk E, Schroder JH, Nanda I, Schmid M, Epplen JT, 1993. High mating success of low rank males in *Limia perugiae* (Pisces: Poeciliidae) as determined by DNA fingerprinting. In: Pena SDJ, Chakraborty R, Epplen JT, Jeffreys AJ, editors. DNA Fingerprinting: State of the Science Progress in Systems and Control Theory Basel: Birkhäuser.
- Travis J, Trexler JC, Mulvey M, 1990. Multiple paternity and its correlates in female *Poecilia latipinna* (Poeciliidae). Copeia:722-729.
- Vega-Trejo R, Backwell PR, 2017. Testing female preferences under more natural conditions: a case study on a fiddler crab. Behavioral Ecology and Sociobiology 71. doi: 10.1007/s00265-017-2314-4.
- Weir LK, Grant JWA, 2010. Courtship rate signals fertility in an externally fertilizing fish. Biology Letters 6:727-731.
- Witte K, Ryan MJ, 2002. Mate choice copying in the sailfin molly, *Poecilia latipinna*, in the wild. Animal Behaviour 63:943-949.

Zane L, Nelson W, Jones A, Avise J, 1999. Microsatellite assessment of multiple paternity in natural populations of a live-bearing fish, *Gambusia holbrooki*. Journal of Evolutionary Biology 12:61-69. **Table S4:** Mean effect size (Hedges' d) estimates for all moderator levels from the nine meta-regression models.

Moderator	Category	n effect sizes	n studies	n species	Mean d	Lower 95% Cl	Upper 95% CI
NA	All	158	58	23	0.583	0.338	0.829
Taxonomic group	Arthropods	33	10	4	0.339	0.040	0.638
	Birds	16	11	3	0.700	0.298	1.101
	Fish	103	34	14	0.625	0.448	0.801
	Mammals	6	3	2	1.459	0.730	2.188
Rate of multiple	High	131	44	15	0.605	0.284	0.926
maung	Low	18	13	5	0.501	-0.018	1.019
Choosing sex	Female	149	50	22	0.578	0.324	0.832
	Male	14	8	9	0.629	0.163	1.095
Design type	Before-and-after	64	28	12	0.787	0.583	0.992
	No pre-test	94	30	15	0.450	0.277	0.623
Model pairing	Manipulated	149	54	22	0.587	0.338	0.837
	Natural	9	5	3	0.561	0.022	1.100
Demonstration type	Affiliation	101	37	17	0.661	0.383	0.939
	Avoidance	2	1	1	1.181	-0.035	2.397
	Courtship	9	3	3	0.708	0.085	1.331
	Mating	46	18	7	0.462	0.147	0.778
Behaviour of focal	Choice frequency	52	19	9	0.563	0.222	0.904
	Reversal frequency	14	9	4	0.403	-0.011	0.817

	Mating frequency	8	3	2	0.426	0.137	0.715	
	Mating latency	4	1	1	0.289	-0.470	1.049	
	Receptive behaviours	5	3	3	0.323	-0.327	0.974	
	Affiliation time	75	39	17	0.753	0.562	0.945	
Generalised or	Generalised	30	11	4	0.692	0.262	1.122	
	Individual	128	49	22	0.530	0.201	0.860	
Animal birth location	Captive	109	38	12	0.609	0.338	0.880	
	Mixed	4	1	1	0.059	-0.922	1.039	
	Wild	45	20	14	0.565	0.235	0.896	



Figure S1. Bubble plot showing the relationship between effect size (Hedges' d) and study year. The dashed line represents the line of fit from a meta-regression including study year as a fixed factor and study identity, species, and phylogeny as random factors (dotted lines represent the 95% confidence interval around the fitted line). The points are scaled according to the inverse of their variance, so that larger points are given greater weight in the model, and represent more reliable estimates.



Figure S2. The relationship between residual effect size and inverse standard error (larger values represent more reliable estimates). The line shows the fitted line from a linear regression of residual effect size against inverse standard error.